Author's response to reviews

Title: Evaluating the effect of innovative motivation and supervision approaches on community health worker performance and retention in Uganda and Mozambique: study protocol for a randomized controlled trial

Authors:

Karin Källander (k.kallander@malariaconsortium.org)
Daniel L Strachan (d.strachan@ucl.ac.uk)
Seyi Soremekun (Seyi.Soremekun@lshtm.ac.uk)
Zelee Hill (zhill.ich@gmail.com)
Raghu Lingam (raghu.lingam@newcastle.ac.uk)
James Tibenderana (james.tibenderana@malariaconsortium.org)
Frida Kasteng (Frida.Kasteng@lshtm.ac.uk)
Anna Vassal (anna.vassal@lshtm.ac.uk)
Sylvia Meek (s.meek@malariaconsortium.org)
Betty Kirkwood (betty.kirkwood@lshtm.ac.uk)

Version: 5
Date: 13 March 2015

Author's response to reviews:

Dear editor(s),

We have revised the manuscript as per the following recommendations:

1. Please explain that informed consent will be obtained from all participants (as opposed to ‘sought’) in the Methods section. - DONE

2. Please include a Discussion section after the Methods section. This can include discussion of any practical or operational issues involved in performing the study, and any other issues linked to the study that do not fall within the previous two headings. - DONE

3. Please move your tables and figure legend section below the reference list. - DONE